Author's response to reviews

Title: Sleep Duration, Snoring Habits and risk of acute myocardial infarction in China population: results of the INTERHEART study

Authors:

Dongfang Xie (xiedongfangpumc@163.com)
Wei Li (liwei@mrbc-nccd.com)
Yang Wang (wangyang@mrbc-nccd.com)
Hongqiu Gu (guhongqiu@mrbc-nccd.com)
Koon Teo (Koon.Teo@phri.ca)
Lisheng Liu (llshypt@yahoo.com.cn)
Salim Yusuf (Salim.Yusuf@phri.ca)

Version: 5 Date: 17 May 2014

Author's response to reviews: see over
Dear editor,

My manuscript “Sleep Duration, Snoring Habits and risk of acute myocardial infarction in China population: results of the INTERHEART study” was revised according to comment. The main modifications are:

1. We have given the full name and affiliation of all the institutional ethics committees in the revised manuscript.
2. We have declared the commercial component of the funding in the Competing Interests section. The full funding statement has moved into the Acknowledgements section.
3. We have used authors' initials in the Authors' Contributions section.

Many thanks for your suggestion. Correspondence and phone calls about this paper should direct to Dongfang Xie at the following address, phone and address:
Address: Division of Biometrics, Fuwai Hospital, Beilishi Road 167, Xicheng district, Beijing, China
Tel: +86 010 88398106,
Fax: +86 010 68331833
E-mail: xiedongfangpumc@163.com

Thanks very much again for your attention to our paper. Once again, thank you for your help to our paper processing.

Yours sincerely,

Dongfang Xie
xiedongfangpumc@163.com